# Gebakken lucht – Identifying Illicit Financial Flows of Oil and Gas Corporations from Africa through the Netherlands<sup>\*</sup>

Denise Brom, Dorian de Gruijter, Joras Ferwerda, and Brigitte Unger

\* The authors note that this is a working paper

'Gebakken lucht' is a Dutch expression that literally means "baked air", proverbially referring to something that appears like quite something, while it is in fact nothing. Oil and gas corporations that delve valuable resources in Africa often have offices in the Netherlands or similar countries with large amounts of money flowing through and little to no real activity. These money flows without underlying substance can be seen as 'gebakken lucht' and are classified as illicit financial flows by UNCTAD. This article shows how one could monitor these immense but hidden illicit financial flows which, when retrieved, would make Africa almost debt-free. We show how red flag indicators can reveal the risk of money laundering, tax evasion, or tax avoidance. We identify 61 different red flags and apply them to five oil and gas corporations (of which one is a 'clean case') active in Angola, Mozambique, Nigeria, and Egypt. These five corporations have their headquarters in Europe with 173 subsidiaries in the Netherlands. Their annual total revenue in Africa is 36.4 billion USD. We demonstrate how such an analysis works and draw preliminary conclusions about the validity of the red flags for monitoring illicit financial flows to reach sustainable development goals.

## 1 Introduction

The role of Multinational Companies (MNCs) in the oil and gas sector in Africa is ambiguous. On the one hand, they are seen as an important driver for development through capital inflows, investment, employment, and the creation of infrastructure. On the other hand, they are perceived as producers of negative externalities such as heavy pollution of air and water, and as responsible for the destruction of the local community and health problems (see eg. Konte and Vincent, 2021). Investments can land in the pockets of corrupt politicians so that the benefits do not reach those who are negatively affected by these companies and who should be compensated for it. <sup>1</sup> This article wants to enlarge the list of negative impacts of MNCs in the oil and gas sector by showing that they do not only take the raw materials out of African countries and leave environmental damage but also do not pay the local taxes they should pay. Developed countries like the Netherlands offer opportunities for through flows of illicit finance. Not much economic activity, not much oil and gas, but only what the Dutch call 'gebakken lucht' – thin air – is being transmitted to shift profits away from Africa. By offering such opportunities to corporations, developed countries are co-responsible for depriving African people of important income.

The estimates of the amounts of unpaid taxes in Africa exceed Foreign Development Aid by far. The UNCTAD's Economic Development in Africa Report of 2020 states that 89 billion USD of illicit financial flows leave Africa, almost the same amount as the 48 billion USD official

<sup>&</sup>lt;sup>1</sup> For a short overview of pros and cons of MNCs in developing countries, see <u>https://vittana.org/19-advantages-and-disadvantages-of-multinational-corporations</u>

development aid and 54 billion USD Foreign Direct Investment together (UNCTAD 2020). Repatriating unpaid taxes would make Africa almost debt-free and is an important task for both developing and involved developed countries. Such tax justice – a pro-poor redistributive tax system aiming to reduce income and social inequality – is a popular and important concept in development theory nowadays. (Cascant-Sempere, 2002: 526)

Adebayo (2021) analyzes the gold mining sector and concludes that the government 'take' - defined as total payments to the government from a mining project (including taxes, fees, and royalties) relative to the mining company's pre-tax net revenue from the same project should be reduced to compensate companies for their investment in local development and their political risks. We think that this is not necessary. On the contrary: we analyze oil and gas companies' tax payment behavior in African countries and conclude that companies reduce the government 'take' themselves - by avoiding paying taxes in African countries. This tax avoidance behavior is enabled by the tax policies of developed countries. Companies can also avoid or minimize the political risks through investment treaties in developed countries, which guarantee their investments in African countries (at least financially) even if there is expropriation by political leaders.

Money laundering, tax evasion, and some forms of tax avoidance are grouped as illicit financial flows by UNC-TAD. Tax avoidance schemes like setting up empty shell companies in tax havens where no business is allowed for foreign companies, with the sole purpose of escaping taxation, are in a grey zone of legality. These tax avoidance schemes use legal loopholes in the international tax system but do not correspond to the spirit of the law. This is why UNCTAD groups them with illegal tax evasion and criminal money laundering as illicit financial flows. There are diverse approaches at the macro-level trying to measure illicit financial flows (see Cobham et al 2021 and Ferwerda and Unger 2021 for an overview and recent results). These methods allow to estimate illicit financial flows between countries and give an estimate of the total size of the problem. Estimates still vary a lot (between 125 billion and 15 trillion USD for the world and between 89 and 200 billion USD for African countries, see Ferwerda and Unger 2021). Cobham et al (2021) see the monitoring of these flows as important for reaching sustainable development goals as defined by UNCTAD.

To monitor illicit financial flows, it is important to assess the risks related to different types of transactions. Illicit financial flows happen in the dark. Therefore, one cannot 'see' and 'monitor' them directly but needs ways to identify them indirectly.

The contribution of this article is to develop a methodology to identify the risks of unpaid taxes and money laundering by MNCs. With the growing resistance to tax avoidance worldwide, it would be valuable to have a risk assessment methodology to indicate which companies engage in illicit tax and money laundering activities. This can help consumers to identify bad social corporate behavior, but also help governments – from developing countries but also the developed countries involved – to focus their efforts on limiting illicit financial flows on the riskiest companies.

To illustrate how such a risk assessment methodology can be set up and applied, this article identifies concrete cases of MNCs tax avoidance, where concrete developed countries are involved and should be urged to close tax loopholes and discuss reparation with the specific companies to reach sustainable development goals.

We apply the methodology to oil and gas companies in four African countries: Angola, Mozambique, Nigeria, and Egypt. We select the Netherlands as the developed country offering opportunities to avoid taxes in Africa. The Corporate Tax Haven list of Tax Justice Network in 2021 lists the Netherlands as number four after the British Virgin Islands, Cayman Islands, and Bermuda (see cthi.taxjustice. net) – hence the first 'non-little island' country on this list. A multitude of double tax treaties and investment treaties create a favorable climate for MNCs to direct their tax avoidance schemes through the Netherlands.

This article focuses on four important corporate groups in the oil and gas industry: Total, BP, Eni, and Saipem. In total these corporations have 137 subsidiaries in the Netherlands. We find in total 35 subsidiaries in the Netherlands related to their activities in Angola, Mozambique, Nigeria, and Egypt. In addition, to test the quality of our risk assessment indicators, we add the Norwegian oil and gas company Equinor as a 'clean case'. By holding the red flags against the selected company cases, this study attempts to determine to what extent the companies are involved in tax avoidance, tax evasion, and money laundering. The red flags function as 'codes' in the analysis of annual company data and aid to give meaning and interpretation to the findings (Boeije, 2016). We disentangle indicators that are measurable with publicly available information and indicators that need further documentation. Our article shows that oil and gas companies can be classified as high-risk, medium-risk, and low-risk companies regarding unpaid taxes. It also shows which risk indicators are most important for identifying illicit financial flows of oil companies. High-interest payments, intragroup loans, complex company structures, and subsidiaries in tax havens are the most telling indicators of oil and gas corporations not paying their fair share of taxes in Africa. While proposed in the literature as indicators for illicit financial flows, characteristics like setting up holdings or negative media attention appear typical for all oil and gas companies and can therefore not function as effective indicators for illicit financial flows in our study. They might qualify as valid indicators though, when applied to the overall economy and compared to other sectors.

This article is organized as follows. After Section 2 describes the risk assessment methodology conceptually, Section 3 gives an overview of relevant indicators for suspect behavior, the so-called red flags, mentioned in the literature, and derives a composite red flag list. Section 4 describes the gathering of data and Section 5 describes the selection of companies and countries. Section 6 reports the results and Section 7 concludes and discusses the contributions of this article.

# 2 Conceptual Framework For A Risk Assessment Methodology For Illicit Financial Flows

Risk assessment is a tool that has been used for decision-making in a wide variety of fields, from health and environment to financial regulation. There is considerable variation in the specific form that the tool takes, reflecting differences in the available data and the specific decisions that it informs. (Rausand, 2011)

The international standard for risk assessment (ISO 31000 and 31010) lists 31 tools and techniques for risk identification, risk analysis, and risk evaluation (ISO 31010: 22), from basic techniques like check-lists and brainstorming to more advanced quantitative tools like Markov and Bayesian analysis. The use of a red flag analysis or composite

indicator analysis to estimate the risks – as previously used for illicit financial flows (see eg. Unger and Ferwerda, 2011, Ferwerda, Deleanu, Unger, 2017, Ferwerda and Kleemans, 2019, Savona and Riccardi, 2017, Riccardi, Milani, Camerini, 2019) – is not recognized yet in the broader risk assessment literature.

This article aims to establish a comprehensive list of indicators that allows identifying companies related to tax avoidance, tax evasion, and money laundering. To establish such a list, we went over the literature from economics, accounting, criminology, and law to find such indicators. We call these indicators red flags. Such a red flag methodology to identify possible company risks is well-known in accountancy to check the quality of financial reports (Robinson and Munter, 2004).

### 2.1 Red flag analysis tool

This article aims to develop a tool that enables its user to determine where the main risks for illicit financial flows are. We are looking for transactions and corporate ownership structures that facilitate illicit financial flows related to tax avoidance and tax evasion. We develop a list of characteristics that are associated with such behavior, the so-called 'red flags'. Although a single characteristic by itself is generally not enough to arouse suspicions, we claim that a combination of these characteristics can.



**Figure 1:** Visual representation of the use of multiple red flags **Source:** Unger and Ferwerda, 2011. This is a visual representation without assumptions on the relative sizes of the relevant groups.

Figure 1 gives a visual representation of this research method. The outer circle represents all the investigated subjects (in this article: oil and gas companies), the circles with stripes and stars are the subgroups found with indicators. The black surface is the congruent of subjects that have the characteristics of all three indicators (for instance an excessively complicated company structure, dividend stripping, and nominee shareholders). The black open circle represents the unknown subgroup we are looking for: MNCs with illicit financial flows. We assume that by using more and more indicators this circle will be filled more and more with darker surfaces.

A priori we do not know the individual effectiveness of the red flags and the (relative) importance of each red flag. Further iterations of the model will have to test the effectiveness of individual red flags (see eg. Unger and Ferwerda 2011 and Ferwerda et al. 2017) and possible interaction effects between different red flags. We assume for now that red flags mentioned in the literature are relevant and have equal importance (Unger and Ferwerda 2011).

Suppose there are two companies (company A and company B, respectively: A, B), and that we have 4 red flags for illicit financial flows (indicated with, in subscript, 1x when the red flag is not present and 1v for when it is). Suppose that after checking for the first three red flags, we determine that both companies have all three of these red flags:

### $\mathsf{A}_{_{1v2v3v}}$ , $\mathsf{B}_{_{1v2v3v}}$

Now suppose that for red flag 4, we find out that it applies to company A, but not to company B:

Then we assume that:

$$R(A_{1y2y3y4y}) > R(B_{1y2y3y4y})$$

With R(A) being the risk or probability that company A is related to illicit financial flows and R(B) being the risk that company B is related to illicit financial flows.

The more red flags are applied, the safer it is to assume that  $R(A_{1v2v3v...Nv}) > R(B1_{v2v3v...Nx})$  with N as the number of red flags applied.

### 2.2 Type I and type II errors

When using indicators, one is always confronted with the trade-off between false positives and false negatives, or the so-called type I and type II errors, respectively. A false positive (or type I error) in this specific research is when a red flag marks a company with a (potential) increased probability of illicit financial flows while it actually is not

engaging in any of such behavior. A false negative (or type II error) occurs when a company related to illicit financial flows does not receive a red flag. Figure 2 visualizes the trade-off between these two errors.



**Figure 2:** Trade-off between Type I and Type II errors **Source:** Unger and Ferwerda, 2011.

Figure 2 is best explained by discussing the two extremes: a very strict indicator and a very broad indicator. A very broad indicator (right end of the graph) marks almost all companies as having this red flag. Therefore, there will be very few false negatives (type II errors; the red, downward sloping line), at the expense of a lot of false positives (type I errors; the blue, upward-sloping line). A very strict indicator (left end of the graph) will mark hardly any object with this red flag. Therefore, there will be almost no false positives (type I errors; the blue line), at the expense of a large number of false negatives (type II errors; the red line). Assuming equal importance, the optimum is the minimum of the sum of these two errors (the green line). Initially, when applying red flags mentioned in the literature, we do not know the amount and types of errors we are making. We assume that companies that are related to illicit financial flows have an increased chance of receiving more red flags than companies that do not (in line with Holland Quaestor, 2020). We can then conclude that the number of false positives (type I errors) will diminish once we focus on the combination of several indicators. Therefore, this approach is better equipped to deal with the false negatives (type II errors) of a single indicator than the false positives (type I errors). We therefore prefer to use relatively broad indicators and accept an increased degree of false positives. (see Unger and Ferwerda 2011)

# 3 Meta-Analysis Of Red Flags For Illicit Financial Flows

The literature on corporate tax avoidance, evasion, and money laundering lists diverse indicators for identifying illicit financial flows of corporations. A test on the reliability of these indicators is often missing. We collect all indicators mentioned in the literature to find out which indicators can be operationalized with existing public data, which ones need non-public data, and which ones cannot be operationalized at all or are redundant.

Our meta-analysis of red flag analyses aims to include as many studies on corporations' illicit financial flows as possible. Some international organizations have already done comparable research. The Financial Action Task Force (FATF) published multiple studies on identifying risky company behavior and the Dutch Center for Research on Multinational Corporations (SOMO) investigates company-specific tax structures (Hietland, 2021). We include the red flag indicators for tax avoidance, tax evasion, or money laundering and categorize them accordingly. In total, our analysis covers approximately 35 different studies and publications. We focus the meta-analysis on the seven most relevant encompassing studies we found on the illicit behavior of companies.

First, this research is inspired by a Profundo study in 2016 that tackles tax avoidance by international companies in the mining industry (Gelder Van et al., 2016). Van Gelder et al. (2016) identify red flags with a literature survey. The study focuses on the role of Dutch holding and finance companies in the mining industry and includes multiple categories of indicators. Some indicators are specifically related to the extractive industry, others to the ownership structure in general. Van Gelder et al. (2016) conclude that the eleven international mining groups match many of their indicators and that the Dutch tax treaty network is relevant for tax avoidance in developing countries.

Second, we include the Tax Integrity Guide of Holland Quaestor, the association for trust offices in the Netherlands. This association can set regulations and guidelines for its members. These guidelines aim to prevent unintended risks for tax evasion and aggressive tax planning. The strong feature of this article is that it categorizes indicators for how seriously they form a risk for tax evasion or avoidance. Some indicators are unconditionally unacceptable, others are only allowed under extra strict supervision and there is a third category of indicators for which contra indicators can apply. The ranking of these different risk levels and potential contra indicators corresponds to the complexity of multinational company structures.

Third, we include indicators for offshore structures by van Koningsveld (2015). Koningsveld (2015) focuses on the legal side of offshore entities and how these can be abused for tax avoidance, financial crimes, and to conceal the Ultimate Beneficial Owner (UBO) of a company. Next to comparing legal forms, the methodology consists of interviews, a literature study, mystery shopping, and case studies. All these methodologies together create a very thorough understanding of the use of offshore companies.

Fourth, the article by the Financial Action Task Force (FAFT) and the Egmont Group (2021) on trade-based money laundering indicators addresses a broad audience of public and private sector professionals for example, in financial institutions, banks, and compliance departments. The indicator categories are structural, trade activities, trade documents, commodities, and transactions. The latter two are almost impossible to identify with the limited information available.

Fifth, we include the Dutch Central Bank's guide "DNB Good Practices for the trust sector" (2019) because of its focus on the process of fiscal research, what an acceptable risks strategy should include, and how to access certain general risks, such as a complex company structure.

The sixth included publication is by Savona & Riccardi (2017) who present in the IARM (Identifying and Assessing the Risk of Money Laundering in Europe) report a country analysis for money laundering risks across business sectors based on quantitative and qualitative research methods. The most vulnerable sectors in the Netherlands are gambling, accommodations, and creative industries. The factors that cause this vulnerability are especially the opaqueness of business ownership and the average profitability of the business.

Finally, FATF & Egmont Group (2018) aim to improve the understanding of government institutions about the risk involved with beneficial ownership structures. Based on over one hundred case studies, the main findings are that shell companies and nominee directors (or informal strawmen) are often used in these structures. A Dutch report lists 6 characteristics of shell companies: 1) a Dutch entity as part of a foreign corporate structure, 2) most transactions are with affiliated entities, 3) few or zero employees in the Netherlands, 4) the Dutch entity is used for financial, legal, or fiscal benefits, 5) financial flows through the entity consist of royalty payments, dividends, or lease payments, and 6) the Dutch entity has relatively high valued assets (ter Haar et al., 2021). Moreover, Ultimate Beneficial Owners are likely to choose a combination of indirect and direct control over their assets. Finally, in money laundering schemes, often a trust office is involved combined with either legal services or an accountant.

The findings of the meta-analysis of these seven publications in combination with expert interviews<sup>2</sup> on how to operationalize and interpret some of the red flags are the foundation of our methodology. These red flags are applied to concrete corporate cases in four African countries. The red flag indicators complement each other and should not be treated separately. As outlined in Section 2, one indicator on its own does not necessarily imply illegitimate behavior, but a combination of indicators increases the risk of illegitimate behavior (Holland Quaestor, 2020). Our list includes 61 indicators in total. The red flags are ordered in six categories along the business process, from how the business is set up, to how the profits are spent and accounted for, and finally how the media reports about the activities carried out. These categories are: 1) corporate structure 2) lack of transparency 3) conspicuous business activities, 4) finance 5) accounting, and finally 6) negative media attention. The list shows that the red flags for tax avoidance and tax evasion sometimes overlap and sometimes stay distinct features.

### 3.1 Corporate structure

The indicators in this category are the following:

- A subsidiary located in a tax haven (van Gelder et al, 2016; Holland Quaestor, 2020; Koningsveld, 2015; IMF, 2007; DNB, 2019; Tavares, 2013; Tax Justice Network, 2015a).
- 2. A Dutch holding or financing company (van Gelder et al., 2016; Actionaid, 2015).
- 3. A Dutch cooperative company (Ministerie van Financiën, 2016; van Gelder, 2016).
- 4. Tax efficiency reasons are indicated for corporate structure (van Gelder, 2016; Holland Quaestor, 2020).
- There is a flow-through company (Holland Quaestor, 2020; FATF & Egmont Group, 2021; DNB, 2019; FATF & Egmont Group, 2018).
- 6. There is a flash entity, set up for one single transaction (Holland Quaestor, 2020; FATF & Egmont Group, 2021).
- 7. A closed limited partnership (besloten commanditaire vennootschap, C.V.) (DNB, 2019).

- The company structure is excessively complicated (Holland Quaestor, 2019; FAFT & Egmont group, 2021; Savona et al., 2017; FAFT & Egmont group; 2018; Holland Quaestor, 2020; FATF & Egmont Group, 2021; DNB, 2019).
- 9. Trade entity is registered in a jurisdiction with weak AML/CFT compliance (FATF & Egmont Group, 2021).
- 10. An entity is registered at a mass registration address (FATF & Egmont Group, 2021).
- 11. There is a lack of effective control mechanisms (Koningsveld, 2015)
- 12. A Special Purpose Entity (SPE) (DNB, 2007; Koningsveld, 2015; Unger, 2007; DNB, 2019).
- 13. Via intermediate holding companies the corporate group could benefit from bilateral tax agreements (Tweede Kamer, 2013; Van Gelder et al., 2016).

An entity can be set up to avoid or evade taxes or launder money. Most red flags that relate to the company structure are relevant for tax avoidance. Entities that are set up for optimizing tax positions and benefiting from tax exemptions using the Netherlands share some characteristics. These are, for example, an excessively complicated structure or mass registration at relatively small office addresses. Flow-through companies hardly need local employees or an actual office because the company only needs the legal existence and domicile in the country of incorporation. Its activities can be conducted abroad by representatives/ employees of the group to which the shell company (legally or economically) belongs. Formal adaptation of contracts only finally have to be conducted by directors that always follow the (verbal) instructions of the group representatives. Another important related indicator is a Special Purpose Entity (SPE), a corporate entity owned by foreign multinationals registered in the Netherlands as a holding or finance company. SPEs are characterized by the fact they can only engage in financial activities and do not have any employees. In 2018, The Netherlands has 12,000 SPEs with a total capital of 3,371 billion euro. 70% of the SPEs use a trust office or other forms of shareholding (DNB, 2020). A study from Garcia-Bernardo et al. (2017) finds that the Netherlands is a conduit offshore financial center (OFC). Conduit OFCs are intermediate countries for financial flows coming from (high tax) origin countries and predominantly sink OFCs as destination. Sink OFCs are typically tax havens, with zero or low corporate tax rates and/or applying favorable tax regimes for specific sources of (foreign) income. Holding companies are a feature of a conduit OFC present in the Netherlands

<sup>&</sup>lt;sup>2</sup> Their knowledge, comments and suggestions to the red flag list and indicators helped to improve, correctly interpret, and operationalize the red flags. Furthermore, sometimes specific questions regarding the application for some of the companies could be clarified. The interviewees were with Martin Bergwerff (international tax consultant), Thom de Jong (Citco group and former DNB supervisor), Joeri de Wilde (Profundo and Investment Strategist at Triodos), Francis Weyzig (Centraal Plan Bureau), and Javier Garcia Bernardo (University of Amsterdam).

### 3.2 Lack of transparency

Ownership concealment indicators are a specific part of the company structure. All these indicators share the common feature that they can be used to hide or conceal the beneficial owner. If there are attempts to hide the ultimate beneficial owner (UBO), this does not necessarily have to do with tax avoidance or other malicious company behavior. Privacy considerations can be a valid argument to hide the beneficial owner.

- 14. The Ultimate Beneficial Owner (UBO) is unknown (Eurodad, 2015; Holland Quaestor, 2020; FATF, 2006; Worldbank, 2011; OESO, 2009; Koningsveld, 2015).
- 15. There is a weak separation of ownership and control (based on interviews for this study).
- 16. There are bearer shares (DNB, 2019)
- 17. A UBO (wrongly) claims tax residency and thereby frustrates the exchange of information. (Holland Quaestor, 2020).
- There are nominee shareholders (Holland Quaestor, 2020; FATF & Egmont Group, 2021; Koningsveld, 2015; Unger, 2006; FATF, 2006; OECD, 2001; DNB, 2019; FATF & Egmont Group, 2018).
- 19. The UBO is a tax resident of a jurisdiction that does not participate in FACTA/CRS regulations. (Holland Quaestor, 2020)
- 20. Recently adjusted ownership structures due to changes in fiscal laws or other causes (DNB, 2019).
- 21. Unrestricted use of legal persons as directors (FATF & Egmont Group, 2018).
- 22. There is a Trust or Company Service Provider (TCSP) acting as director (van Gelder, 2016; Koningsveld, 2016; FATF & Egmont Group, 2018)
- 23. The name of the trade entity is a copy/very similar to a well-known corporation (FATF & Egmont Group, 2021).
- 24. There is a STAK (Stichting Administratie Kantoor) in the company structure, this increases the complexity of the corporate structure (DNB, 2019).

If the UBO is not registered or publicly known, there is a lack of transparency. The weak separation between ownership and control is related to unlisted companies. If there are no external shareholders, there is less external control, accountability, and transparency. Company misbehavior is therefore more likely to go unnoticed. Bearer shares and nominee shareholding are related to the ownership structure. Bearer shares, where the owner of a share is identified via certificates, are not allowed anymore in the Netherlands. Nominee shareholding separates the legal and economic ownership. A service provider can hold shares on behalf of a client and become the legal owner. It is not easy for authorities to recognize this and find the UBO. Moreover, the company structure can be further complicated by using many persons as legal directors and by adjusting the company structure as a result of changes in fiscal laws. Finally, the jurisdiction where the UBO is registered influences the information process.

### 3.3 Conspicuous business activities

The following red flags are related to business activities:

- 25. Business activity of an entity is not appropriate for the type of address (FATF & Egmont Group, 2021).
- 26. A business entity lacks an online presence, or online presence suggests other activities than stated in reports (FATF & Egmont Group, 2021).
- 27. The business entity shows a noticeable lack of business activity (FATF & Egmont Group, 2021).
- 28. The entity has unexplained periods of inactivity (FATF & Egmont Group, 2021).
- 29. The trade activity of an entity is not in line with the stated line of business (FATF & Egmont Group, 2021).
- 30. The trade entity engages in transactions or shipping routes that are inconsistent with standard business practices (FATF & Egmont Group, 2021).
- 31. There are unreasonable low-profit margins in the transactions (FATF & Egmont Group, 2021).
- 32. A new trade entity engages in high-volume or value activities (FATF & Egmont Group, 2021).
- 33. The entity purchases commodities that exceed the economic capacities of the entity (FATF & Egmont Group, 2021).
- 34. The price of oil and gas is undervalued (van Gelder et al., 2016; Holland Quaestor, 2020; FATF & Egmont Group, 2021).
- 35. A limited number of employees (van Gelder et al., 2016; Holland Quaestor, 2020; FATF & Egmont Group, 2021).

- 36. Payment for commodities is made by another entity without economic reason (FATF & Egmont Group, 2021).
- 37. A company is not in compliance with regular business obligations (FATF & Egmont Group, 2021; Tax Justice Network, 2015).

These indicators aim to find business entities or business behavior that are unusual and therefore can be suspicious. Some indicators are related to a deviation from the stated line of business. For example, an address registration not appropriate for the activities, unusual shipping routes or transactions for the sector, or when a business entity suddenly engages in the trade of very different products. Other indicators are related to a lack of business activities. For example, lacking online presence, lack of payroll transactions compared to the number of employees, an unexplained period without any activities, or a very lowprofit margin. Finally, it can be suspicious if a business suddenly enters a market with high entry barriers or invests/ purchases commodities exceeding the economic capacity.

#### 3.4 Finance

The category 'financial flows and payments' contains a mix of indicators related to tax avoidance, tax evasion, and money laundering:

- 38. There are payments for patents to a jurisdiction with a lower profit tax (Eurodad, 2015; Holland Quaestor, 2020).
- 39. There are royalty payments to a jurisdiction with a lower profit tax (Weyzig, 2013; Holland Quaestor, 2020).
- 40. There are intra group loans and payments between entities of the same corporate group. (Weyzig, 2013; Actionaid, 2015; Holland Quaestor, 2020; Koningsveld, 2015).
- 41. Dividend payments do not follow hierarchical lines but are redirected (European Commission, 2016).
- 42. Dividend stripping (Holland Quaestor, 2020).
- 43. Management fee payments to jurisdictions with a low-profit tax (Van Gelder et al., 2016).
- 44. Cash deposits are consistently just below reporting thresholds (FATF & Egmont Group, 2021).
- 45. Large round payments in a sector where this is unusual (FATF & Egmont Group, 2021).

- 46. Payments go in circles (FATF & Egmont Group, 2021).
- 47. There is a high value of money remittance (FATF, 2016).
- 48. Payments to local governments (Curtis, 2015; TRACE International, 2016; Savona et al., 2017).
- 49. There are advanced tax rulings in developing countries (van Gelder et al., 2016).

Payments for patents or royalty payments to a jurisdiction with a low-profit tax are a well-known way to lower tax payments in a high tax jurisdiction. Similar to this is the use of intra-group loans and payments resulting in interest payments between different entities belonging to the same corporate structure. Intra-group loans may be used to shift costs to a high tax country and therefore lower the taxable amount of profit. Management fee payments can be used similarly.

Dividend payments are redirected when dividend payments in the corporate structure do not follow hierarchical lines but are redirected via intermediate financial holding entities, in countries as the Netherlands, which combines a participation exemption with a beneficial network of tax treaties. This can indicate tax avoidance.

Dividend stripping is not legal anymore in the Netherlands. The most recent important legal adjustment is from January 2020. However, it could still be useful to investigate if this happened in previous years. This is characterized by selling or lending shares as soon as a dividend payment is announced and buying them back after the dividend has been paid. The dividend to be received is included in the selling price and, after the dividend has been paid, the shares are repurchased at a lower price. The result is that the dividend has effectively not been paid and no dividend tax is owed, or dividend tax is paid at an excessively low rate.

### 3.5 Accounting

The fifth category contains the indicators for fiscal or bookkeeping 'tricks'.

- Underreporting of exports (Le Billion, 2011; Kar & Spanjers, 2015; Weyzig, 2013; Curtis, 2015; Curtis, 2013; OSISA, Third World Network Africa, Tax Justice Network Africa, Action Aid International, and Christian Aid, 2009; FATF & Egmont Group, 2021).
- 51. Overreporting of imports (OSISA, Third World Network Africa, Tax Justice Network Africa, Action Aid International, and Christian Aid, 2009; Kar & Spanjers, 2015; Curtis, 2015; Weyzig, 2013).

- 52. No country-by-country reporting where this is mandatory (OSISA, Third World Network Africa, Tax Justice Network Africa, Action Aid International, and Christian Aid, 2009; Curtis, 2015; Anglo American, 2016; Holland Quaestor, 2020).
- 53. Accelerated depreciation of assets (OSISA, Third World Network Africa, Tax Justice Network Africa, Action Aid International, and Christian Aid, 2009).
- 54. An entity has relatively many immaterial assets (van Gelder et al., 2016).
- 55. There is a difference between the reported and taxed income (Frank, et al., 2009).
- 56. Tax returns in developing counties are nicely rounded amounts (Jaskiewicz, 2015; Holland Quaestor, 2020).
- 57. Contracts regarding complex trade transactions seem unusually simple. (FATF & Egmont Group, 2021).
- 58. Contracts or invoices have vague descriptions of commodities or documents are missing. (FATF & Egmont Group, 2021).
- 59. Last-minute changes in the payment details of a transaction (FATF & Egmont Group, 2021).
- 60. The owner or manager evades or avoids taxes (Chyz, 2013).

Underreporting exports and overreporting imports are ways to shift profits to countries with a lower tax rate and thus reduce the total amount of paid taxes. Indicators related to (immaterial) assets are also used to reduce the tax burden. On the other hand, indicators in contract irregularities are suspicious of money laundering. Such as when contracts are too simple for the transaction, for example, internet copies, or when rejected contracts or tax fills are resubmitted. Finally, it is suspicious if a company is not in line with regular business obligations such as filing VAT or country-by-country- reporting.

### 3.6 Negative media attention

61. The company received negative media attention (van Gelder et al., 2016; Holland Quaestor, 2020; FATF & Egmont Group, 2021).

If the company receives negative media attention, this can be a signal something is wrong with the company. This of course depends on the content of the news articles.

Table 1 gives a visual overview of the red flags in their respective categories. Appendix I provides additional details for some red flags.

### 4 Data

We use the annual reports of the selected corporate groups retrieved from Company.Info<sup>3</sup> and the Dutch Chamber of Commerce<sup>4</sup> and their ownership structures from the Orbis database to operationalize the red flags, determine whether there are links between the four African countries and the Netherlands, and visualize the group structures. This study mainly uses annual reports of the selected corporate groups from 2018, in combination with earlier reports, because for most companies these are the latest available documents at the time of our research. We analyze the balance sheet, profit, and loss accounts, and cash flow statements in the annual reports of the Dutch subsidiaries to find annual revenues, profits, tax paid, intra-group loans, and other relevant financial information. We distinguish capital structures and money flows between entities within the corporate group. Furthermore, qualitative content analysis (Bryman, 2016) of the annual reports of the Dutch subsidiaries provides information on the activities of the companies to check for red flags. <sup>5</sup>

# 5 Selection And Description Of Corporations And Countries

### 5.1 Selection of corporations

Our selection procedure for relevant cases consists of several steps. In the first step, we select international oil and gas companies operating in African developing countries with a link to the Netherlands. For this, we use the Orbis dataset.<sup>6</sup> The selection process is based on the following criteria:

<sup>&</sup>lt;sup>3</sup> Company.Info is part of the FD Mediagroup and gives access to the annual reports and financial data of all Dutch companies. More information on: <u>https://companyinfo.nl</u>

<sup>&</sup>lt;sup>4</sup> Dutch Company Register. More information on: <u>https://www.kvk.</u> <u>nl/zoeken/handelsregister/</u>

<sup>&</sup>lt;sup>5</sup> As this is a partly interpretative qualitative study developed in an unexplored field, the study is automatically susceptible to researcher bias. Often there are multiple explanations for the existence of red flags, and one red flag on its own is not sufficient to prove tax planning. But analyzing each corporate group by means of red flags gives a consistent focus and quantifiable way of defining the risk on tax avoidance. The validity is verified by combining different sources, backed up by interviews with experts in the field. Moreover, to reduce selection bias, countries and companies are selected based on objective criteria.

<sup>&</sup>lt;sup>6</sup> Orbis provides information on almost 400 companies and entities worldwide and provides comparable financial data and extensive corporate ownership structures. The database is owned by Bureau van Dijk, which became Moody's Analytics company in 2017. Although Orbis data on developing countries is limited, data on companies operating in the Netherlands is more extensive. More info on Orbis dataset can be found on: <u>https://www.bvdinfo.com/ en-gb/our-products/data/international/orbis</u>

Company structure		Lack of transparency		Accounting	
Subsidiary located in a tax haven	-	Unknown UBO	F	Underreporting expoxt	
Dutch holding	-	Weak separation	i 🐂 👘	Overreporting imports	lie e l
Dutch cooperative	l 🛏 👘	Bearer shares	🛏	No country-by-country report	l in in l
Tax efficiency	i 🗖	Tax residency	l 💌	Accelerated depreciation	i 🖻 📜
Flow-through companies	i- r	Nominee shareholders		<u>Many immaterial assets</u>	
<u>Flash entity</u>		<u>UBO tax resident - non</u> FACTA/CRS_	-	Difference reported/taxed income	-
Bilateral tax agreements	-	Adjusted ownership	<b>–</b>	Rounded tax return	<b>•</b>
A closed limited partnership (C.V.)		Unrestricted directors	-	developing country Unusual simple contracts	-
Complicated structures		Trust company		Issues with contacts and invoices	-
Weak AML/CFT jurisdiction	i=	Similar name	-	Late changes in payments	
Mass registration address	-	A STAK		<u>Owner/manager</u> avoids taxes	
Lack supervision/ control	-	Conspicuous business activity		Finance	]
A SPE		Unappropriate type of addres	<b>•</b>	Payments for patents	] 🖻
Negative media	1	Lack online presence		Royalty payments	<b>•</b>
Negative media attention		Lack business activity	-	Intragroup loans	
	1	Unexplained inactivity	<b>–</b>	Redirected dividend payments	
		Not in line with stated business	i 🗖	Dividend stripping	i =
		<u>Non-standard transactions/</u> routes	<b>F</b>	Management fee	<b>-</b>
		Very low profitmargines		Cash deposit below threshold	<b>—</b>
		High volumes for new enitiy	<b>F</b>	Unusual, large round payments	-
		Exceeding economic capacity.	<b>F</b>	Payments in circles	<b>•</b>
		Few employees	<b>•</b>	Money remittance	<b>•</b>
		<u>Undervaluing oil and gas</u> <u>prices</u>		Advance tax rulings developed country	-
		<u>Not in compliance businesss</u> <u>obligations</u> Payment by other entity		Payments to local government	

#### Table 1: Categorized indicators for illicit financial flows

**Source:** Made by the authors. Underlining the name of the indicator = the indicator cannot be found in public data. Yellow flag = indicator for tax avoidance, orange flag = indicator for tax evasion, red flag = indicator for money laundering

- The company is operating in African developing countries where the gas and oil industry sector accounts for a significant percentage of Gross Domestic Product (GDP)
- The company has connections with the Netherlands, but does not have significant operating or exploration activities in the Netherlands (e.g. no headquarter)
- The company has publicly available lists of involved subsidiaries
- There is sufficient financial information available

• The companies provide yearly annual reports and data are available in the Orbis dataset.

We select the following international oil and gas companies based on these criteria: Total SE, BP P.I.C., Eni S.p.A., and Saipem S.p.A (Table 2).<sup>7</sup> In total 21 companies that have subsidiaries in both the Netherlands and Africa are found in Orbis. Out of this list Saipem, BP and Total are selected, because these companies are active in Angola,

<sup>&</sup>lt;sup>7</sup> The selected companies only include European multinationals. Chinese, Russian and American multinational companies are also active in Africa, but generally provide less financial information and are therefore not included in this study.

Nigeria, Egypt, or Mozambique. Other companies like Shell or Exxon Mobile are not included, as these companies have large operating activities in the Netherlands as well and as a result do not classify as conduit companies. Total is the fourth largest international oil and gas major with a presence in more than 130 countries, and the largest integrated major<sup>8</sup> in Africa, with as main producing countries Angola and Nigeria. Total is headquartered in Paris, France. BP is on the list of largest international oil and gas companies as well with operations in 33 countries worldwide. Operating revenues in Africa amounted to \$7.2 billion in 2018. BP is headquartered in London, United Kingdom. ENI S.p.A. is relevant because it came up in several corruption scandals and is active in Africa since 1954. ENI is headquartered in Rome, Italy. ENI is active in 68 countries worldwide and has been a key player in Northern Africa for over sixty years (Villa, 2016). The group plays a large role in the new gas project in Mozambique. Saipem S.p.A is a service provider in the oil- and gas industry and present in 60 countries worldwide. The headquarter is located in Milan, Italy. Saipem is a leading company in engineering, drilling and construction in the energy and infrastructure sector. These four companies are compared to a clean company case: Equinor ASA. This is a Norwegian oil company headquartered in Stavanger. According to our survey sent to NGOs involved in taxes and developing countries and interviews with some colleagues, Equinor was seen as the oil company with the highest corporate social responsibility.9 Equinor is present in Angola and Nigeria.

The 5 corporate groups have 173 subsidiaries in the Netherlands. Their total revenue in Africa amounts to \$36.4

billion annually. For a more comprehensive description of the corporate groups see appendix III.

### 5.2 Selection of countries

The second step consists of the selection of African countries where the four multinational companies are operating. The selection process is based on the following criteria:

- The developing country has a large amount of proven oil or gas reserves
- It appears in the top 20 of the world's ranking of proven reserves

This resulted in the selection of the following countries: Angola, Mozambique, Egypt, and Nigeria (see Table 3)<sup>10</sup>. We find 35 Dutch subsidiaries involved in either subsidiaries or permanent establishments in the selected countries. As Table 3 shows, Angola and Mozambique do not have double tax arrangements with the Netherlands, while Egypt and Nigeria have had them for more than twenty years. They together with Mozambique have bilateral investment treaties (BITs) with the Netherlands. These treaties usually reduce the political risk of investment and insure investment in a developing country. DTAs usually allow favorable tax arrangements (van Gelder et al., 2016). Investigating countries that do not have a DTA with the Netherlands can control for the effects of DTAs and help to find out if there are other reasons to settle in the Netherlands as well.

Source: Retrieved from annual reports. Revenue in 2018. Amounts in USD.

<sup>&</sup>lt;sup>8</sup> An integrated oil and gas company is a business entity that engages in the exploration, production, refinement, and distribution of oil and gas, as opposed to companies that specialize in just one segment.

<sup>&</sup>lt;sup>9</sup> Appendix III provides an extensive overview of all Dutch subsidiaries of the selected companies.

Dutch subsidiaries Revenue in Africa Corporate group Headquarter France 60 12.6 billion Total SE United Kingdom 29 7.2 billion BP P.I.C. 44 15.3 billion Italy ENI S.p.A. 9 3.04 billion Saipem S.p.A. Italy 32 5.23 billion Norway Equinor AsA 173 36.37 billion Total Sample

<sup>&</sup>lt;sup>10</sup> Other African countries that are not included in the scope of this study could be relevant as well. However, the selected companies are mainly active in the Angola, Mozambique, Egypt and Nigeria.

**Table 2:** Selection of international oil and gas companies.

	Angola	Mozambique	Egypt	Nigeria
Double Tax Agreement	No	No	20-05-2000	09-12-1999
Bilateral investment treaty	No	01-09-2004	01-03-1998	01-02-1994
Crude annual oil production (2018) bln barrels <sup>1</sup>	0.52	-	0.23	0.70
Oil reserves in bln barrels (world ranking) <sup>2</sup>	7.8 (16)	-	3.3 (24)	36.9 (10)
Gas production (2018) bln cubic feet <sup>1</sup>	222.9	182.4	2258.8	1682.0
Gas reserves bln cubic feet (world ranking) <sup>2</sup>	12,113 (37)	100,000 (13)	63,000 (16)	203,444 (10)

Number of subsidiaries (more than 50% ownership; excluding permanent establishments)

Total SE		3		11
BP P.I.C.		3	3	6
Eni S.p.A.		1	5	4
Saipem S.p.A	3	1	1	5
Equinor ASA				5

#### Table 3: Details selected countries

1: Data from International Energy Statistics (IEA, 2018) – numbers are converted from Ktoe to billion barrels/billion cubic feet. 2: Data from US Energy Information Administration (EIA, 2021). The world rankings date from 2019.

The sample includes the two largest oil producing countries of Sub-Saharan Africa, Nigeria and Angola. Nigeria holds 36.9 billion barrels of proven crude oil reserves, compared to 7.8 billion in Angola. In both countries the government plays a large role in regulating the oil industry. In Nigeria companies can operate under a license in the form of joint ventures between the federal government and either an international oil company or a sole risk operator. Companies can also operate under product sharing agreements (PSAs) or risk service agreements (RSAs). In Angola the right to produce and explore oil or gas is granted by licenses or concession agreements. The state is the owner of all natural resources, and mineral rights are exclusively assigned to the National Concessionaire (ANPG), which can associate with other entities through corporation, consortium, PSAs or RSAs. In Nigeria oil companies are subject to income tax of 65.75 percent in the first five years and 85 percent after five years. Contractors in PSAs are subject to a tax rate of 50 percent under the Petroleum Profits Tax. In Angola all entities that carry out oil operations are subject to Petroleum Revenue Tax of 50 percent for PSAs or 65,75 percent for partnerships and RSAs (EY,2019). Exemptions and specific taxation rules are possible (Ferreira & Cunha, 2021).

Mozambique and Egypt are known for their large gas reserves. Mozambique is not producing crude oil yet, but is one of the largest natural gas reserve holders in Africa with estimations up to 100 trillion cubic feet. In comparion, gas reserves in Africa are estimated to be 615 trillion cubic feet (Rocha \$ Mendes, 2021; EIA, 2021). Egypt holds 63 trillion cubic feet of gas reserves and 3.3 billion barrels of oil reserves. In Mozambique the gas sector is in a very early stage with two well-established fields in the South of country and a recently discovered field in the Rovuma Basin (2010). Production is expected to start in 2023/2024 (Rocha & Mendes, 2021). In Egypt the oil and gas sector accounts for 15 percent of GDP. In both Mozambique and Egypt, the natural resources are owned by the state. In Mozambique the government role is not fully developed yet. In Egypt the state can award oil and gas exploration rights by virtue of concession agreements or PSAs. Companies pay an income tax of 32 percent in Mozambique, and 40,55 percent in Egypt (EY, 2019).

For a more comprehensive overview of main activities and regulations in the four African countries, see appendix II.

# 6 Results On The Red Flags Of The Selected Corporations

### 6.1 Corporate structure

All five corporate groups have subsidiaries in the Netherlands, ranging from nine for Saipem to sixty Dutch subsidiaries for Total. However, only for Total, BP, Eni, and Saipem some of these subsidiaries are related to the four selected African countries. For Equinor no relationship is found.

These subsidiaries have stakes in subsidiaries in the African countries or organize their activities through permanent establishments. For Total, Eni, and BP the activities in Angola are managed by Dutch subsidiaries with permanent establishments in Angola. Total also has Dutch subsidiaries with permanent establishments in Egypt and Mozambique, and BP has Dutch subsidiaries with permanent establishments in Egypt as well. Before 2012, it was fiscally attractive to set up a Dutch subsidiary with a foreign permanent establishment instead of a foreign subsidiary, because foreign losses of permanent establishments could be credited with Dutch profits (until 2012 due to object exemption).

In addition, the Dutch subsidiaries are located at mass registration addresses (Appendix V) and it seems that there is a lack of effective control mechanisms as most subsidiaries are managed by the same people. On the other hand, most of the subsidiaries only report financial assets and are empty shells. Take Total, the revenues flowing through Dutch subsidiaries exceed \$7.2 billion in 2018. For ENI \$6.1 billion of revenues were flowing through Dutch subsidiaries. A large amount is related to activities in developing countries. The subsidiaries with permanent establishments in Africa do report tangible and intangible assets but do not have any activities in the Netherlands.

Moreover, Total, BP, and ENI report subsidiaries in sink-countries. Saipem and Equinor, however, do not use conduit companies in the Netherlands to channel profits to sink countries. Saipem has some subsidiaries in Luxembourg (classified as conduit OFC). The organization structures of Total, BP, and ENI are excessively complicated as well (appendix IV for organization structures). The distance between the ultimate owner and Dutch or African subsidiaries goes up to seven levels for Total, and 5 levels for BP and ENI. Equinor has a simple structure. It manages the activities in Angola directly via Norwegian subsidiaries in Nigeria are managed by Nigerian subsidiaries.

### 6.2 Lack of transparency

All five oil and gas companies show a lack of transparency as it is hard to find the UBO.<sup>11</sup> Companies listed on the stock market have a registry of who owns these shares, but the UBO is hard to determine since these shares are owned by many other companies and trusts.

Only for Total SE the involvement of trust companies is found, out of which one is managed by a Stichting Administratiekantoor (STAK). Stogg Eagle Funding B.V. has Stichting STOGG Eagle Funding as shareholder and IQ EQ Management B.V. as director. The annual reports do not show why Stogg Eagle Funding B.V. was created, only loans from Total Holdings Nederland B.V. to this company are found but the beneficiary is unclear. The annual report of Total reports that Stogg Eagle Funding B.V. has operations in Nigeria, but provides no further information. Stogg Eagle Funding B.V. likely functions as a flow-through company. Brass Holdings B.V. is another subsidiary of Total that seems to function as a flow-through company.

### 6.3 Conspicuous business activities

This study was unable to identify most of the conspicuous business activities from public data. However, it stands out that most of the subsidiaries have a limited number of employees. For Total, we find that all subsidiaries have zero employees, except for Total E&P Nederlands B.V. (578 employees). BP does not have employees in the Dutch subsidiaries related to operations in Africa. ENI has 47 employees in Eni International B.V. and zero employees in the other subsidiaries. Saipem has 8 employees in Saipem International B.V. and 12 in Saipem Finance International B.V. and Equinor Holding Netherlands B.V. has 14 employees.

### 6.4 Finance

Total, BP, ENI, and Saipem report many intragroup loans between Dutch entities to entities worldwide. For Total, many loans are found from Total Holdings Netherlands B.V. to Nigeria and one to Angola. BP reports large values of finance receivables in the reports of BP Egypt Production B.V., BP Netherlands Upstream B.V., and BP Angola (Block 18) B.V. For Eni loans from Eni International B.V. in the Netherlands to subsidiaries in Bermuda and the United Arab Emirates via international bank corporations are found. These loans are used to control activities in Angola and Mozambique. Saipem uses Finance International B.V.

<sup>&</sup>lt;sup>11</sup> At the time of writing this article (July 27, 2022), the deadline to register the UBO in the Dutch company register (KvK) has not passed yet. We analysed a sample of subsidiaries for all of the five oil and gas corporations (on July 27, 2022) and found that Saipem and Equinor registered the UBO of their subsidiaries, but Total, BP, and ENI registered the UBO of only some of their subsidiaries.

in the Netherlands to finance activities of other entities in the corporate group. For Equinor intragroup loans are registered as well, but none of these are related to the selected African countries.

For ENI service payments between ENI Angola Production B.V. in the Netherlands to Angola LNG Limited in Bermuda are found as well. This indicates that untaxed incomes are shifted to Bermuda and the effective tax rate of Eni Angola Production B.V. is lowered due to higher costs. Angola LNG Limited is also partly owned by Total, BP, and Chevron, which likely means that these corporate groups are all shifting the related profits of the LNG project to Bermuda.

Total, BP, ENI, and Saipem seem to benefit from tax and/or investment treaties by setting up subsidiaries in the Netherlands. As a result, dividend payments do not follow hierarchical lines but are redirected through multiple Dutch subsidiaries before they end up with the parent company. ENI also profits from a tax treaty between the Netherlands and the United Arab Emirates. ENI's activities in Mozambique are partly financed through a company in the United Arab Emirates. This is a red flag because (future) profits are shifted to the United Arab Emirates, corporate income taxes in the United Arab Emirates are zero and the DTA between the United Arab Emirates and the Netherlands can be used to avoid withholding taxes as well. Under this treaty, the withholding tax on dividends paid to United Arab Emirates entities owned by the government is zero percent and five percent for non-government entities, and the withholding tax on interest and royalties is zero percent (EY, 2021). Debt financing in Mozambique lowers the effective tax rate in the country and government revenues as a result. The holding company in the United Arab Emirates is fifty percent owned by Exxon Mobile, so the reports of Exxon Mobile should be considered as well.

### 6.5 Accounting

Many of the indicators in this category could not be identified with the public data available. As a result, it is unclear if companies over- or under-report assets or accelerate the depreciation of assets. What the annual reports do reveal is that the Dutch subsidiaries of all five corporate groups have large amounts of immaterial assets. In addition, for Total, BP, and ENI differences between reported and taxed income are found. The Dutch subsidiaries report large amounts of deferred taxes. Eni International B.V. has been making losses for years in a row and is not paying any taxes in the Netherlands. According to the annual report of 2018, the effective tax rate in 2019 was 0.9 percent, compared to 0.6 percent in 2018. Fiscal losses of operations are carried forward for nine years via interest and operating income. Although we do not find large amounts of deferred taxes for Saipem S.p.A. the annual reports show zero effective tax rates for Saipem International B.V. It did not pay income taxes in the Netherlands in 2014, 2015, 2018, and 2019 because profits were offset against losses. Snamprogetti Netherlands B.V. (a subsidiary of Saipem S.p.A.) did not pay income taxes in 2014 and 2015 and stated in the annual reports that it did not expect profits in the years to come.

Remarkable of the activities of ENI related to Mozambique is that Eni Mozambique LNG Holding B.V attributes the

	TOTAL	BP	ENI	SAIPEM	EQUINOR
CORPORATE STRUCTU	IRE				
SUBSIDIARY IN TAX HAVEN	Subsidiaries in Bermuda, The Cayman Islands, Mauritius Island, Panama, Hong Kong, and Cyprus	Subsidiaries in the British Virgin Islands, Bahamas, Mauritius, Barbados, and Luxembourg	Subsidiaries in Bermuda, British Virgin Islands, Cyprus, Cayman Islands, Bahamas and Luxembourg, and the United Arab Emirates	No subsidiaries in tax havens. Only 2 in Luxembourg.	No subsidiaries in tax havens.
DUTCH HOLDING	60 Dutch subsidiaries/ 14 (indirectly) related to selected countries	29 Dutch subsidiaries/ 8 (indirectly) related to selected countries	44 Dutch subsidiaries/ 9 (indirectly) related to selected countries	9 Dutch subsidiaries/ 4 (indirectly) related to selected countries	32 Dutch subsidiaries/ none are related to selected countries
FLOW-THROUGH COMPANIES	Likely for Stogg Eagle Funding B.V. and possibly for Brass Holdings B.V.				

COMPLICATED	BO-distance of 7	BO-distance of 5	BO-distance of 5	Rather simple structure,	Rather simple
STRUCTURES	levels is found.	levels is found.	levels is found. Especially, complex structures with regards to activities in Mozambique and Angola	BO-distance of 3 levels is found.	structure, BO- distance of 1 or 2 levels for activities in Nigeria and Angola.
MASS REGISTRATION ADRESS	Yes: Bordewijklaan 18, The Hague	Yes: d'Arcyweg 75, Europoort Rotterdam	Yes: Strawinskylaan 1725 or Strawinskylaan 1727, Amsterdam	Yes: Strawinskylaan 1647 Amsterdam or Westfrankelandsedijk 7 Schiedam	Yes: Weena 760, Rotterdam
A SPECIAL PURPOSE ENTITY	Most Dutch subsidiaries are SPEs	Most Dutch subsidiaries are SPEs	Most Dutch subsidiaries are SPEs	Most Dutch subsidiaries are SPEs	Most Dutch subsidiaries are SPEs
BILATERAL TAX AGREEMENTS	For activities in Nigeria and Egypt	For activities in Egypt. for Nigeria it is unclear as not many activities have been found.	For activities in Nigeria and Egypt	For activities in Nigeria and Egypt	No use has been made of a treaty between Netherlands and Nigeria
LACK OF TRANSPA	RENCY				
UNKNOWN ULTIMATE BENEFICIAL OWNER UBO	No full registration of UBO	No full registration of UBO	No full registration of UBO	No full registration of UBO	No full registration of UBO
STRONG SEPARATION	Listed company with minor shares	Listed company with minor shares	Listed company, but Italian government holds 30,33 percent of shares	Listed company, but Eni S.p.A. holds 30,5 percent of the shares-	Listed company, but Norwegian government holds 67 percent of shares.
NOMINEE SHAREHOLDERS	Might be the case for Stogg Eagle Funding B.V. and Brass Holdings B.V.	JPMorgan Chase Bank (holds 25 percent of shares) allows for nominee shareholdings			
TRUST COMPANY	Stogg Eagle Funding B.V. and Brass Holdings B.V.				
STICHTING ADMINISTRATIEKANTOOR (STAK)	Stichting Stogg Eagle Funding				

### CONSPICUOUS BUSINESS ACTIVITIES

#### FEW EMPLOYEES

578 employees Zero employees in in Total E&P the Netherlands Nederland B.V, other subsidiaries have zero employees 47 employees in Eni International B.V., other subsidiaries have zero employees

8 employees in the Netherlands in Saipem International B.V. and 12 in Saipem Finance International B.V. 14 employees in Equinor Holding Netherlands BV

### FINANCE

INTRAGROUP LOANS	Many intra group loans from Total Holdings Netherlands B.V. (NL) to especially Nigeria and one to Angola	Large values of finance receivables in the annual reports of BP Egypt Production B.V. (2018), BP Netherlands Upstream B.V. (2015), and BP Angola (Block 18) B.V. (2018)	Loans from Eni International B.V. via international bank corporations to subsidiaries in Bermuda and the United Arab Emirates controlling activities in Angola and Mozambique	Loans from Saipem Finance International B.V. (NL) to other entities in the corporate group	Intragroup Ioans, but none related to Nigeria or Angola
REDIRECTED DIVIDEND	Dividends are transferred through multiple Dutch subsidiaries and end up in France with the parent	Dividends from subsidiaries in Egypt are transferred through Dutch subsidiaries and US subsidiaries and end up in the UK with the parent. Dividends are also transferred through multiple Dutch subsidiaries.	Dividends from subsidiaries in Mozambique, Egypt, and Nigeria are transferred through Dutch subsidiaries and end up in Italy with the parent.	Dividends from subsidiaries in Angola, Nigeria, and Egypt are transferred through Saipem International B.V. and end up in Italy with the parent.	
MANAGEMENT FEE			Eni Angola Production B.V. pays for services to Angola LNG Limited in Bermuda		
ACCOUNTING					
MANY IMMATERIAL ASSETS	Balance sheet of Dutch companies mostly consists of financial assets and receivables	Balance sheet of Dutch companies mostly consists of financial assets and receivables	Balance sheet of Dutch companies mostly consists of financial assets and receivables	Balance sheet of Dutch companies mostly consists of financial assets and receivables	Balance sheet of Dutch companies mostly consists of financial assets and receivables
DIFFERENCE REPORTED/TAXED INCOME	Large amounts of deferred taxes	Large amounts of deferred taxes	Large amounts of deferred taxes		
OWNER/MANAGER AVOIDS TAXES	The use of Total Management B.V.	The use of BP Management Netherlands B.V. and PB Management International B.V.			

#### **NEGATIVE MEDIA ATTENTION**

NEGATIVE MEDIA ATTENTION 1,031 negative news stories in ORBIS

2,463 negative news stories in ORBIS 353 negative news stories in ORBIS and legal proceedings relating to Nigerian Agip Exploration Ltd.

Total

RP

FNI

169 negative news stories in ORBIS and legal proceedings related to Snamprogetti S.p.A with activities in Nigeria

Sainem

Fauinor

721 negative news stories in ORBIS

Table 4: Company details of relevant red flags

	10101	Di	LIVI	Suipeni	Lyunioi
Number of identified tax avoidance indicators	16	14	15³	11	8
Number of identified tax evasion indicators	8²	6²	6	3	3
Number of unidentified indicators	15	15	16	17	13
Source: calculation by the authors.					

Table 5: Overview of the number of red flags per company

Indicator	Total	BP	Eni	Saipem	Equinor	Most relevant for:
Subsidiary located in tax haven	+	+	+	-	-	Avoidance
Excessively complicated structures	+	+	+	-	-	Avoidance and Evasion
Nominee shareholders	+	+	-	-	-	Avoidance and Evasion
Intra-group loans and interest payments	+	+	+	+	-	Avoidance and Evasion
Redirected dividend payments	+	+	+	+	-	Avoidance
With intermediate companies benefit from tax agreements	+	+	+	+	-	Avoidance
Difference between reported and taxed income	+	+	+	?	-	Evasion
TCSP acting as director	+	-	-	-	-	Avoidance and
						Evasion
Flow-through companies	+	-	-	-	-	Evasion
A STAK (Stichting Administratie Kantoor)	+	-	-	-	-	Avoidance
Management fee payments to low tax jurisdiction	-	-	+	-	-	Avoidance

**Table 6:** Distinctive red flags for the 5 companies selected.**Source:** Made by the authors.

losses to Eni International B.V., while the latter is merely a flow-through company of loans and interests. Eni Mozambique LNG Holding B.V. joined the fiscal unity of Eni International B.V. in 2017 just after activities in the Coral South in Mozambique were started. By attributing the expenses for activities in the Coral South to Eni Mozambique LNG Holding, the profitable base is lowered which leads to profit erosion in the Netherlands.

### 6.6 Negative media attention

The five corporate groups are mentioned in negative news stories in Orbis, ranging from 2463 negative stories for BP to 721 negative stories for Equinor. Saipem is only negatively mentioned 169 times and ENI 353 times.

For the four corporate groups, eleven to sixteen red flag indicators for tax avoidance are found and this indicates a high risk of tax avoidance for the 35 Dutch subsidiaries (Table 4). Several reasons for the incorporation of the companies in the Netherlands are DTAs, the Dutch participation exemption, and the fiscal benefits of foreign permanent establishments (until 2012). For the clean case, only 8 indicators for tax avoidance and 3 indicators for tax evasion are found. This suggests that these indicators might be too general and apply to companies regardless of tax avoidance behavior. In addition, eleven distinctive red flags indicate a high risk of tax avoidance or tax evasion (Table 5). These red flags do not show up for the clean case. Some of the indicators that are most important to indicate tax avoidance are the use of international treaties, special financial institutions, and intragroup loans. Next to these findings this study also gives some insight into which indicators can be operationalized with public data, non-public data, and indicators that are hard to operationalize to begin with (Table 6).

# 7 Conclusions, Recommendations, And Caveats

As Table 6 shows, the selected companies show different degrees of risk of illicit financial flows. We identified 14-16 red flags of tax avoidance for Total, BP, and Eni, classifying them as corporations with a high risk of tax avoidance, followed by the medium-risk corporation Saipem with 11 red flags and our low-risk clean case Equinor with 8 red flags. With regard to tax evasion, a similar picture emerges. 6-8 red flags for high risk of tax evasion of Total, BP, and Eni, compared to 3 red flags of Saipem and Equinor.

We analyzed red flags relating to 1. corporate structure 2. lack of transparency 3. conspicuous business activities 4. finance 5. accounting, and 6. negative media attention. In Table 8 we show which of the 61 red flag indicators from

the literature can be found in public data for free or can at least be bought from commercial suppliers like ORBIS and company.net. Unfortunately, only less than half of the red flags (30 out of 61) can be found in public databases (either paid or for free). There is public data on corporate structures, but very little on lack of transparency and almost none on conspicuous business activities of corporations. As Table 1 shows, many of the red flag indicators for conspicuous business activities apply to organized crime and money laundering. They require private company information or confidential data from tax authorities or other government agencies.

Also, many indicators on finance and accounting need non-public information on bank transactions, or from tax authorities and auditors. Some indicators (still) cannot be operationalized. There are efforts made between researchers, the customs, and tax authorities to operationalize and measure underreporting and overreporting of exports and imports. Similar the over- or undervaluation of the gas and oil price could be traced. Payments to local politicians would need information from the bookkeepers of the company or police information.

Nevertheless, we found that oil corporations behave conspicuously in all six categories for the 19 indicators applicable.

Regarding the first category, corporate structure, all five cases have Dutch subsidiaries, which follows evidently from our selection of cases. However, the company structures of the high-risk cases show many subsidiaries in typical tax havens like the Bermudas, British Virgin Islands, Mauritius, Cayman Islands, and Luxembourg. The medium-risk case Saipem has only two subsidiaries in Luxembourg, while the clean case Equinor has none. Moreover, the high-risk cases show very complicated corporate structures with 5-7 organization layers, compared to three layers for the medium-risk case and only two layers for low-risk case Equinor.

Regarding lack of transparency, finding the ultimate beneficial owner of the Dutch subsidiaries is a challenge. Even our clean case Equinor shows a lack of transparency about the ultimate beneficial owner. Having a transparent register of beneficial ownership to find out who owns which company is essential for monitoring illicit financial flows, and the oil industry as a whole consisting of stock market listed corporations does not comply with this criterion. Since March 28 2022 all companies in the Netherlands should have registered the ultimate beneficial owner in the company registry of the Dutch Chamber of Commerce. Till today, after this deadline, many subsidiaries of all our selected oil and gas corporations in the Netherlands have still not registered the ultimate beneficial owner.

Regarding finance, intra-group loans seem typical for the oil and gas sector, while complex dividend payment struc-

tures seem to be an important red flag. Dividends are transferred between multiple Dutch subsidiaries before they end up with the parent company in a foreign country. For both the high and medium-risk companies we find intragroup loans to subsidiaries in the selected developing countries. In some cases, these loans are also transferred through tax havens, like Bermuda or the United Arab Emirates. For the clean case, we see no loans to subsidiaries in either Nigeria or Angola.

Bilateral tax treaties seem to play an important role in illicit financial flows. While the risky companies use tax treaties for activities in Nigeria and Egypt, the clean case Equinor doesn't make use of tax treaties between the Netherlands and Nigeria. For ENI it is relevant to mention that the subsidiaries in the Netherlands and the United Arab Emirates were set up to finance activities in Mozambique, thanks to the bilateral tax treaty between the Netherlands and the United Arab Emirates.

Accounting characteristics show that a high volume of deferred taxes is typical for high-risk companies while they are not for the medium and low-risk ones.

Negative media attention seems to be the curse of oil and gas companies. They are in the media for ecological reasons, for concerns of public health, and some for tax and money laundering reasons. There were in total around 5000 negative news stories in the Orbis database.<sup>12</sup> The largest amounts for BP followed by Total. But also our clean case Equinor got 721 negative news stories in the ORBIS database, for example relating to tremendous losses due to its operations and management flops in the US in 2020. (www.newsunenglish.no, May 11, 2020) Equinor has more negative news stories in ORBIS than Saipem and Eni. Negative news seems to be a too general indicator to identify IFFs and can therefore be misleading without further in-depth analysis.

Our study confirms many of the usual suspects of indicators like having subsidiaries in tax havens, having no business activities in the country where 'gebakken lucht' is flowing through and the gas seems to have evaporated. Encompassing monitoring would, however, red flag the whole oil industry. All corporations follow the law of doing Country by Country Reporting, so this red flag became redundant through international law (for a more critical view on how this is being done in practice, see www.taxjusticenetwork.com).

#### Recommendations

How could our study be used by governments and international organizations? First, we demonstrate that it is possible to identify and monitor corporations' tax behavior systematically. We showed that some indicators (e.g. payments to local governments) cannot be operationalized for now, other indicators (e.g. money remittances and advanced tax rulings) need confidential data, and some indicators are redundant (e.g. indicators about SPEs, through flow companies, flash entities, and STAK are overlapping to some extent and could be merged) or misleading (e.g. negative news referring to non-tax issues). This means that one could start monitoring oil and gas corporations with only a few red flag indicators from public sources which reduces the amount of work needed. International organizations could collect red flag information for a World Matrix (see Table 7). The rest of the 61 indicators would need privacy-protected data for monitoring and should be done by public authorities. Second, we show that misbehavior is appearing in four African countries done by four corporations, of which three – Total, BP, and Eni – stick out as a particularly high risk. Third, with this type of analysis governments of developed countries could take steps to negotiate directly with the relevant corporations to reduce misbehavior and take their corporate social responsibility seriously. Fourth, developed countries could take legal steps to reduce tax loopholes, like those hidden in bilateral tax treaties. Fifth, developing countries could identify how their taxes disappear into through flow tax havens and could make this an issue.

Conspicuous behavior of high-risk corporations can be detected by creating a monitoring matrix for indicators like subsidiaries in tax havens, complicated structures, intragroup loans, redirected dividends, bilateral tax agreements, and differences between reported and taxed income. Oil corporations share features that would become red flags if one would expand the analysis to all sectors, like the lack of economic activity in a country and the lack of transparency about the beneficial owner of the company. Tax Justice Network traces these features at a macro level for countries. We think that one should also do this at the micro level for corporations, and at the industry level. In addition, an index of negative appearances in newspapers could be added. An overall index of Social Corporate Responsibility could then be composed which should include the contribution and damage of corporations to the economy, public tax revenue, the environment, labor relations, and health.

Creating awareness of the enormous amounts of money that are taken out of developing countries, being able to calculate the amounts and to identify the perpetrators of illicit financial flows both at a micro and meso level, seems an important step to bringing tax money from developed countries and tax havens back to developing countries.

There is still a way to go to establish an encompassing framework that covers all corporations and all countries. But with this handmade example of five corporations in four African countries, we hope to have shown that moni-

<sup>&</sup>lt;sup>12</sup> Based on the Orbis "number of negative news stories since last year" search function for the 5 corporations (Total, ENI, BP, Saipem, and Equinor) in 2021.

Indicato	r	Free public data / paid public data	Non-public data	Comment
Corporat	te structure	public data		
1.	Subsidiary located in tax haven	Corporate annual report / ORBIS		
2.	Dutch holding	Corporate annual report / ORBIS / Company.Info		
3.	Dutch cooperative	Chamber of Commerce / Company.Info		
4.	Tax efficiency	Corporate annual report		
5.	Flow-through companies	Company.Info / Chamber of Commerce		
6.	Flash entity		Chamber of Commerce combined with Bank account data	
7.	Closed limited partnership (C.V.)	Chamber of Commerce		
8.	Complicated structures	ORBIS		
9.	Weak AML/CFT jurisdiction	Corporate annual report / ORBIS		
10.	Mass registration address	Corporate annual report / ORBIS / Compony.Info		Incidentally possible to find in leaks
11.	Lack supervision/control			Multidimensional indicator without a central registry
12.	SPE	Company.Info/Chamber of Commerce		
13.	Bilateral tax agreements	Corporate annual report		
Lack of t	ransparency			
14.	Unknown UBO	Chamber of Commerce		Incidentally possible to find in leaks
15.	Weak separation	Corporate annual report		
16.	Bearer shares			No central registry of bearer shares
17.	Tax residency		Tax authority data	Several tax authorities might need to be consulted

18.	Nominee shareholders		The applicable trust	
			office should know this	
19.	UBO tax resident non		Tax authority data	Several tax authorities
	FACTA/CRS			might need to be
				consulted
20.	Adjusted ownership	Chamber of Commerce		
	Unrestricted directors	Company.Info/ Chamber of		
		Commerce		
22	Trust company	Chamber of Commerce		
	Similar name			
23.	Similar name	Corporate annual report /		
		Chamber of Commerce		
	STAK	Chamber of Commerce		
Conspicu	ious business activities			
25.	Inappropriate type of	Chamber of Commerce and		
	address	land registry		
26.	Lack online presence	Internet		
27.	Lack business activity		Company.Info/ Chamber	
			of Commerce combined	
			with Bank account data	
28.	Unexplained inactivity		Bank account data	
29.	Not in line with stated	Chamber of Commerce		
	business			
30.	Non-standard		Bank account data/	
	transactions/routes		Customs data	
21	Very low profit margins		Private company	
51.	very low projit margins		information	
32.	High volumes for new		Bank account data	
	entity			
33.	Exceeding economic		Bank account data	
	capacity			
34.	Undervaluing oil and gas		Customs data with	
	prices		additional information on	
			regular prices	
35.	Few employees	Company.Info/ Chamber of		
		Commerce		
36.	Payment by other entity		Bank account data	
	Not in compliance		Reports of Tax authorities	
	business obligations		/ auditors	
L			,	

Finance				
	Payments for patents	Company.Info/ Chamber of		
	,	Commerce		
39.	Royalty payments	Company.Info/ Chamber of Commerce		
40.	Intragroup loans	Company.Info/ Chamber of		
41.	Redirected dividend	Commerce Company.Info/ Chamber of Commerce		
42.	Dividend stripping		Stock market and tax authorities	
43.	Management fee		Auditors	Incidentally in annual reports
44.	Cash deposit below threshold		Bank account data	
45.	Unusual large round payments		Bank account data	
46.	Payments in circles		Bank account data	
47.	Money remittance		Money transfer offices	
48.	Payments to local government			Hard to trace, especially when done in cash
49.	Advance tax rulings developed country		Tax authorities	
Accounti			L	I
50.	Underreporting export		Customs data with additional information on regular prices	
51.	Overreporting imports		Customs data with additional information on regular prices	
52.	No country-by-country	Corporate annual report or		
	report	CbCr documentation		
53.	Accelerated depreciation		Tax authorities or auditors	
54.	Many immaterial assets	Corporate annual report		
55.	Difference reported/taxed income		Tax authorities combined with Chamber of Commerce	

56.	Rounded-off tax return developing country		Tax authority of developing country	
57.	Unusual simple contracts		Private company information	
58.	Issues with contracts and invoices		Private company information	
59.	Late changes in payments		Private company information	
60.	Owner/manager avoids taxes		Tax authorities	Incidentally possible to find in leaks
Negative	e media			
61.	Negative media attention	Internet / ORBIS		Incidentally possible to find in leaks

#### Table 7: Sources of indicators, separated by data availability

**Source:** Interpretation of the authors. This table provides an overview of our experience with the relevant sources of information to operationalize the indicators. The categories can be interpreted as follows. Public data: data that can be retrieved through annual reports of the corporate entities or public data available from the Dutch Chamber of Commerce. Source in indent if (some) payment is needed. Non-public data: data where more insight into tax administration, bank data, accounting data, or police data is needed.

toring illicit financial flows is feasible also at a micro-level. This allows to identify new ways of corporations' restructuring, and change in behavior, and to identify high-risk sectors, which macro-indicators might miss otherwise.

#### Caveats

A problem with the use of annual company reports is the internal reliability and validity of the research. As the contents of the annual reports differ per corporate group and per company, not the same data can be analyzed for each company and data can be missing. Many indicators cannot be identified with the public information we used. Only 30 out of 61 red flag indicators could in principle be identified with publicly available data. Access to data from the Tax Offices in the Netherlands, the police, the Dutch Central Bank on transactions, and the developing countries is necessary for a more comprehensive view of tax evasion and money laundering by oil and gas companies. This data is necessary to check the reliability of the indicators as well. As a result, the main focus of this study is the corporate structure of the selected oil and gas companies and possible financial flows between Dutch entities and extractive operations in the selected developing countries. Physical flows, export, and trade data are not included in the scope of this study. The red flags only indicate unusual company behavior and can point at suspect corporations but cannot conclude anything about the exact amount of tax avoidance. This would need further investigation by the tax authorities. So, our red flag analysis is only a first step to identifying and pointing out unusual corporate behavior.

This research focuses on five corporate groups in the oil and gas industry and cannot generalize the findings to other corporate groups in the same industry. Micro-level analysis is needed to provide statements on the tax planning behavior of other corporate groups. However, if similar tax planning strategies are found for each of the four corporations, other corporations in the industry will likely use similar techniques. The red flag list is generalizable and can be used in further research and for other industries and sectors, as it is based on sources that are not limited to the context of the oil and gas industry.

The authors wish to thank for their input to this study: Martin Bergwerff, Rick van Dijk, Javier Garcia-Bernardo, Thom de Jong, Francis Weyzig, Joeri de Wilde, Jan van Koningsveld, and Michael Getzner. This article has been prepared in the Spatial and Network Analysis of Tax Evasion (SPAN) project, financed by infobox Crimineel en Onverklaard Vermogen (iCOV), a Dutch cooperation network of National Police, Tax Office, Customs, Financial Police, Central Judicial Collection Agency, Financial Intelligence Unit, special law enforcement agencies, the Public Prosecutors Office, Authority for Consumers and Markets and the Dutch Central Bank.

### References

- Adebayo E. (2021) Is conventional wisdom about resource taxation correct? Mining evidence from transparency reporting – World Development 2021 volume 146
- Actionaid (2015) An Extractive Affair: How one Australian mining company's tax dealings are costing the world's poorest country millions.
- Aertsen, A.E. De Trust, Beschouwingen over invoering van de trust in Nederland, Kluwer.
- Anglo American (2016), Tax and Economic Contribution Report 2015: Driving Change, Defining Our Future.
- Cascant-Sempere, M. J. (2022). Grounding ActionAid's Tax Justice Campaigns in Nigeria. Development and Change, 53(3), 525-550.
- Chyz, J. A. (2013). Personally tax aggressive executives and corporate tax sheltering. Journal of Accounting and Economics, 56(2), 311-328.
- Cobham, A., J Garcia-Bernardo, P. Jansky and M. Palansky (2021) Statistical Measurement of Illicit Financial Flows in Sustainable Development Goals: Tax Avoidance by Multinational Corporations Working Papers IES, 2021/24 Charles University Prague, Faculty of Social Sciences, Institute of Economic Studies, revised Jul 2021.
- Curtis, M. (2013). Malawi's Mining Opportunity: Increasing Revenues, Improving Legislation.
- Curtis, M. (2015). Improving South Africa's mining revenues and transparency: The need for government action.
- De Nederlandsche Bank (2007). Bijzondere Financiele Instellingen- integriteitsrisico bij grote geldstromen, DNB Kwartaalbericht maart 2007, p. 65-68.
- De Nederlandsche Bank (2019). Good Practices Fiscale Integriteitsrisico's voor trustkantoren 2019.
- Financial Action Task Force (2001). Report on Money Laundering Typologies 2000-2001, Paris.
- Financial Action Task Force (2002). Review of the FATF Forty Recommendations, Consultation paper.
- Financial Action Task Force (2006). The Misuse of Corporate Vehicles, including Trust and Company Service Providers: Typologies Report, Paris.
- Financial Action Task Force (2016). Money or Value Transfer Services. Paris: Financial Action Task Force- Organization for Economic Cooperation and Development.
- Financial Action Task Force & Egmont Group (2018). Concealment of Beneficial Ownership. Retrieved via: https://www.fatf-gafi.org/ media/fatf/documents/reports/FATF-Egmont-Concealment-beneficial-ownership.pdf

- Financial Action Task Force & Egmont Group (2021). Trade-Based Money Laundering Risk Indicators. Retrieved via : https://www. fatf-gafi.org/media/fatf/content/images/Trade-Based-Money-Laundering-Risk-Indicators.pdf
- Ferwerda, J. and B. Unger (2021), How Big Are Illicit Financial Flows? The Hot Phase of IFF Estimations, in: Unger, B., Rossel L. and J. Ferwerda (2021, eds), Combating Fiscal Fraud and Empowering Regulators, Oxford University Press, UK, Chapter 5
- **Eurodad** (2015) Fifty Shades of Tax Dodging: The EU's role in supporting an unjust global tax system.
- European Commission (2016) Commission Staff Working Document COM(2016) 23 final.
- Frank, M., Lynch, L., Rego, S. (2009) Tax reporting aggressiveness and its relation to aggressive financial reporting. The Accounting Review.
- Haar, B. van, Bergwerff, M., de Haan, A., Moore, P., Vording, H. and Weyzig, F. (2021). Op weg naar acceptabele doorstroom.
- Hietland, M. (2021). Keep watching. The tax avoidance structures of ViacomCBS.
- Holland Quaestor (2020). Tax Integrity Richtlijn van Holland Quaestor leden.
- Hull, J. (2012). Risk management and financial institutions (Vol. 733). John Wiley & Sons.
- IMF (2007). Concept of Offshore financial Centres: In search of an Operational definition, april 2007.
- Institute of South Africa (OSISA), Third World Network Africa, Tax Justice Network Africa, Action Aid International and Christian Aid (2009) Breaking the curse: How transparent Taxation and fair taxes Can Turn Africa's Mineral Wealth into Development.
- Jaskiewicz, D. (2015) Tax evasion 'red flags' can act as warning of greater fraud, Blackhawk Intelligen.
- Kar, D. & Spanjers J. (2015) Illicit Financial Flows from Developing Countries: 2004-2013, Global Financial Integrity.
- Koningsveld, T. J. van, (2015). De offshore wereld ontmaskerd: Een studie naar aard en omvang van het misbruik maken van offshore vennootschappen door Nederlandse (rechts) personen. Kerckebosch.
- Konte, M., & Vincent, R. C. (2021). Mining and quality of public services: The role of local governance and decentralization. World Development, 140, 105350.
- Le Billion, P. (2011) Extractive sectors and illicit financial flows: What role for revenue governance initiatives?

Ministerie van Financiën (2016, September 20), "Internationaal fiscaal (verdrags)beleid- Brief van de Staatssecretaris van Financiën", Tweede Kamer, vergaderjaar 2016–2017, 25 087, nr. 131.

- OECD (2001). Report 'Behind the corporate Veil', Using corporate entities for illicit purposes; Paris.
- OECD, (2009). Money laundering Awareness handbook for tax Examiners and Tax Auditors.
- Rausand, M. (2013). Risk assessment: theory, methods, and applications (Vol. 115). John Wiley & Sons.
- Savona, E. U., & Riccardi, M. (2017). Assessing the risk of money laundering in Europe. Final report of project IARM.
- Tavares, R. (2013). Relationship between Money Laundering, Tax
   Evasion and Tax Havens (Thematic Paper on Money Laundering).
   Bruxelles: European Parliament- Special Committee on Organised
   Crime, Corruption and Money Laundering.
- Tax Justice Network. (2015a). Financial Secrecy Index 2015- Final results. Tax Justice Network. Retrieved from http://www.financialsecrecyindex.com/PDF/FSI-Rankings-2015.pdf
- Tax Justice Network. (2015b). Financial Secrecy Index 2015- Methodology. Chesham: Tax Justice Network. Retrieved from http://www. financialsecrecyindex.com/PDF/FSI-Methodology.pdf

TRACE International (2016), Global Enforcement Report 2015.

- Tweede Kamer (2013), "Evaluatie van eventueel verdragsmisbruik; BEPS action 6; treaty abuse and treaty shopping is one of the most important concerns of the BEPS", Tweede Kamer, vergaderjaar 2013-2014, 25 087, nr. 61
- Van Gelder, J. W., De Wilde, J., van Koningsveld, J., & Ferwerda, J. (2016). Tax avoidance by mining companies in developing countries-An analysis of potential Dutch policy initiatives. Amsterdam: Profundo.
- Villa, M. (2016). In for the Long Haul: Italy's Energy Interests in Northern Africa. ISPI In for the Long Haul: Italy's Energy Interests in Northern Africa | ISPI (ispionline.it)
- Weyzig, F. (2013) Taxation and development: effects of Dutch tax policy on taxation of multinationals in developing countries.
- Worldbank (2011) Van der Does de Willebois e.a., The Puppet Masters, Stolen Asset Recovery Initiative
- UNCTAD (2020), Africa could gain 89 billion USD annually by curbing illicit financial flows', UNCTAD United Nations Conference on Trade and Development.
- Unger, B. Siegel, e.a. (2006) The amounts and effects of Money laundering, Utrecht School of Economics, in opdracht van het Ministerie van Financiën.

- Unger, B (2007). The Scale and Impacts of Money Laundering, Edward Elgar Publising Limited, UK.
- Unger, B., & Ferwerda, J. (2011). Money Laundering in the Real Estate Sector. Edward Elgar.